22-Year magnetic solar cycle [Hale cycle] responsible for significant underestimation of the Sun's role in global warming but ignored in climate science

Martijn van Mensvoort

Key Points:

1

2

3

4

5

6

7 8	•	22-year Hale cycle solar minima show for the period 1890-1985 a high SST solar sensitivity (1,143 $^{\circ}{\rm C}$ per W/m2)
9	•	22-year Hale cycle temperature profile amplitude (0,215 $^{\circ}$ C) is higher than for the
10		11-year Schwabe cycle $(0,122$ °C)
11	•	Solar influence on climate is underestimated without a 22-year Hale cycle temper-
12		ature correction ($\sim 0.1 \text{ W/m2}$)

Corresponding author: Martijn van Mensvoort, martijn.van.mensvoort@gmail.com

13 Abstract

Reconstructions for global temperature development show an upward oscillation for the 14 period of the 1880s through 1980s. This oscillation is being associated with natural vari-15 ability and the temperature rise between the 1910s and 1940s with increased solar ac-16 tivity. The temperature impact of the 11-year solar cycle [Schwabe cycle] and the phys-17 ical mechanism involved are insufficiently understood. Here, for the 22-year magnetic 18 solar cycle [Hale cycle] a seawater surface temperature [SST] impact is described of 0,215 19 $^{\circ}$ C (0,238 ± 0,05 $^{\circ}$ C per W/m2); the derived impact for the 11-year cycle is 0,122 $^{\circ}$ C 20 $(0.135 \pm 0.03 \text{ °C per W/m2})$. Also, a parallel development is described for seawater sur-21 face temperature [HadSST3 dataset] and the minima of total solar irradiance [LISIRD 22 dataset] after a correction based on the 22-year solar cycle polarity change. With the cor-23 rection, the combination of the positive and negative minima shows for the period 1890-24 1985 a high SST solar sensitivity: $1,143 \pm 0,23$ °C per W/m2 (with 90,5% declared vari-25 ance). This implies that the Sun has caused a warming of 1,07 °C between Maunder min-26 imum (late 17th century) and the most recent solar minimum year 2017 - which is well 27 over half of the intermediate temperature rise of approximately 1,5 °C. The results demon-28 strate that the 22-year cycle forms a crucial factor required for better understanding the 29 Sun-temperature relation. Ignoring the 22-year cycle leads to significant underestima-30 tion of the Sun's influence in climate change combined with an overestimation of the im-31 pact of anthropogenic factors and greenhouse gases such as CO2. 32

³³ Plain Language Summary

Global temperature development shows an upward oscillation for the 1880s through 34 1980s. This oscillation is associated with natural variability: increased solar activity largely 35 explains the temperature rise between the 1910s and 1940s. However, the temperature 36 impact of the 11-year solar cycle is insufficiently understood. Here, for the 22-year mag-37 netic solar cycle a seawater surface temperature impact is described of 0.215 °C, while 38 the derived impact for the 11-year cycle is only 0,122 °C. Also, a parallel development 39 is described for seawater surface temperature and the minima of total solar irradiance, 40 after a correction based on the 22-year solar cycle polarity change. With this correction, 41 the combination of the positive and negative minima shows for the period 1890-1985 a 42 high solar sensitivity for seawater surface temperature: 1,143 °C per W/m2. This also 43 implies that the Sun caused a warming of 1,07 °C between Maunder minimum and so-44 lar minimum year 2017, well over half of the intermediate temperature rise of approx-45 imately 1,5 °C. The 22-year cycle forms a crucial factor for better understanding the Sun-46 temperature relation. Ignoring the 22-year cycle leads to underestimation of the Sun's 47 influence in climate change (+ overestimation of anthropogenic factors and greenhouse 48 gases such as CO2). 49

50 1 Introduction

In a 2006 Dutch scientific report by the Royal Netherlands Meteorological Insti-51 tute (KNMI) in collaboration with the NIOZ, is reported that prior to 1950 the influ-52 ence of humans on temperature had been negligible (de Jager et al., 2006). This makes 53 the period prior to 1950 ideally suited for studying the influence of the Sun on temper-54 ature. In the current research, the influence of the Sun on seawater surface temperature 55 is being studied for the period 1890-1985. This time frame includes 3 periods in which 56 the temperature trend has changed direction plus it includes a total of 10 solar minimum 57 years. According to experts, prior to 1880 insufficient data is available for a reliable es-58 timate of the global seawater surface temperature; only after the year 1950 the uncer-59 tainty margin decreases to a low level for most regions of the world (Smith & Reynolds, 60 2003). Among experts there is consensus that the heat content of the ocean system is 61 probably the best indicator of global warming (Cheng et al., 2019); logically, the warm-62

ing of the seawater surface temperature is therefore probably a more relevant indicator
 than the warming of the atmosphere. In this study the HadSST3 dataset is used for sea-

⁶⁵ water surface temperature.

There is controversy about the solar influence on climate on a wide range of aspects. 66 Estimates for the temperature effect of the 11-year solar cycle [Schwabe cycle] vary from 67 less than 0.05 °C (barely recordable) (de Jager et al., 2006) to more than 0.25 °C (Camp 68 & Tung, 2007). However, a much larger temperature effect is expected for the same amount 69 of energy when it involves a much longer timespan. For a 200-year cycle, the temper-70 71 ature effect is 2 to 4 times larger than for the 11-year cycle, particularly due to accumulation of energy within the ocean system (de Jager et al., 2006); for even longer periods 72 the impact can be 5 to 10 times larger (Shaviv, 2005, 2012). Solar sensitivity (= the tem-73 perature response to solar activity) represents a complicated phenomenon because times 74 scale and solar cycle phase is required to be taken into account. Additionally, the im-75 pact of a solar amplification factor (of unknown size) for the TSI signal measured at the 76 top of the atmosphere should also be taken into consideration. 77

The controversy also concerns the share of the Sun in the 0.8 °C warming in the 78 20th century: available estimates range from 7% (0,056 °C) to 44-64% (0,35-0,51 °C) 79 (Scafetta, 2013). The compilation method of the historical dataset for total solar irra-80 diance [TSI] is an important part of the controversy as well (Solanki et al., 2013). Since 81 the 1990s, even the scientific legitimacy has been debated in relation to the compilation 82 method used by different research groups involved; among experts this issue is known 83 as the ACRIM-PMOD controversy (Scafetta et al., 2019). Large opinion differences have 84 arisen with regard to the TSI construction method. The widely adopted method of Lean 85 (Lean et al., 1995) is based on just 2 magnetic components and produces a curve which 86 shows the highest TSI values in the late 1950s. While, for example, the method of Hoyt 87 & Schatten (Hoyt & Schatten, 1993) is based on 5 magnetic components and produces 88 a curve which shows the highest TSI values near the beginning of the 21st century. This 89 means that estimates for the influence of the Sun on the climate differ both numerically 90 and fundamentally to a great extent; numerically, the controversy involves impact dif-91 ferences of nearly a factor of 10. 92

In climate science the influence of the Sun is studied, among other things, by means 93 of the 11-year solar cycle. However, fundamentally, it has been established since the be-94 ginning of the 20th century that the 22-year magnetic solar cycle [Hale cycle] forms the 95 origin of the 11-year sunspotscycle (Hale, 1908). This is important because two consec-96 utive 11-year cycles exhibit structural differences; an illustrative example for this involves 97 the Gnevyshev-Ohl rule (Zolotova & Ponyavin, 2015), which relates to the number of 98 sunspots between 2 consecutive maximums. It is therefore remarkable that the 22-year 99 cycle is hardly taken into consideration in climate science. IPCC reports do not men-100 tion the existence of the 22-year Hale cycle (Hiyahara et al., 2008). Descriptions else-101 where in the scientific literature indicate that manifestations of the 22-year cycle are be-102 ing presumed to be not sensitive to the polarity change; however, the foundation for such 103 assumptions is unclear. Because, for example, in 2008 it has been determined that since 104 Maunder minimum the coldest phase of the 22-year cycle takes place (under the influ-105 ence of cosmic rays) during the minima that occur when the polarity is positive; the mag-106 netic solar poles are then located in their original position (IPCC, 2013; Hiyahara et al., 107 2008).108

This study therefore distinguishes two categories of solar minima (Mursula & Hiltula, 2003): (1) 'positive minima' [P], which arise during the phase when the magnetic polarity of the northern solar hemisphere is positive with both poles in the original position; and (2) the 'negative minima' [N], which arise during the phase when the magnetic polarity of the northern solar hemisphere is negative with switched positions for both poles. This is crucial because solar radiative forcing trend analysis is usually based on solar minimum years because the phase of the solar cycle must be taken into account (in order

to avoid effects that origin from phases differences in the solar cycle). Solar sensitivity 116 is typically higher for long term perspectives; therefore especially for periods much longer 117 than the 11/22-year solar cycle the phase of the solar cycle needs to be considered in or-118 der to separate trend effects due to the 11/22-year solar cycle from trend effect in longer 119 term perspectives. This is explained by the fact that minima are both "more stable" and 120 "more relevant" than maxima (IPCC, 2013). IPCC AR5 presents a definition for the TSI 121 which refers only to the minima. In terms of the physical processes involved this is ex-122 plained by the fact that the number of sunspots and solar flares is relatively small dur-123 ing the minima. Both represent the two magnetic components in the Lean method, which 124 also represent the basis of the LISIRD TSI dataset used in this study. The maxima are 125 accompanied by relatively large fluctuations, which exhibit higher uncertainty than the 126 minima. This is because the result at the maxima depends more strongly on the mag-127 netic components used in the reconstruction (Lean et al., 1995; Hoyt & Schatten, 1993). 128 This explains the fundamental relevance of the choice made in this study to use the per-129 spective of the solar minimum years as the most important point of reference for study-130 ing the climate impact of the 22-year magnetic solar cycle. Additionally, the sunspot con-131 vention is used in order to separate also two categories of solar maxima: (1) 'even max-132 ima' [E] from (2) 'odd maxima' [O] (Ross & Chaplin, 2019). Figure 1 describes the Hale 133 cycle based on Wilcox Solar Observatory data; the LISIRD TSI dataset is added for ref-134 135 erence at the bottom.



Key: Lt.Solid - North; Dashed - -South; Med.Solid - Average: (N-S)/2; Hvy.Solid - Smoothed Average

Figure 1. The amplitude of the poloidal solar magnetic field is largest during the years around the TSI minima; Wilcox Solar Observatory data shows that the field changes polarity during the TSI maxima (source: (WSO: Solar Polar Field Strength [.gif] http://wso.stanford.edu/gifs/Polar.gif)). The Lt.Solid (blue) and Dashed (red) graphs show activity of the magnetic north pole and inverted south pole, respectively; the Med.Solid (black) graph represents the average magnetic activity and the Hvy.Solid (bold black) graph represents the smoothed average. The LISIRD TSI is added at the bottom.

¹³⁶ 2 Materials and Methods

The materials used in this study involve datasets for global sea surface temperature and total solar irradiance. For global sea surface temperature is used the Hadley

Centre Sea Surface Temperature dataset [HadSST3 : https://www.metoffice.gov.uk/ 139 hadobs/hadsst3/data/download.html (MetOffice, 2020)] presented by the Hadley Cen-140 tre Met Office, who's sea surface temperature datasets serve in IPCC AR5 (IPCC, 2013). 141 For total solar irradiance is used the Lasp Interactive Solar IRadiance Datacenter dataset: 142 Historical Total Solar Irradiance Reconstruction, Time Series [LISIRD: http://lasp 143 .colorado.edu/lisird/data/historical_tsi/ (Kopp, 2019)], which is an unofficial 144 dataset presented by LASP principal investigator Dr. Greg Kopp. On Greg Kopp's TSI 145 Page the LISIRD is being described to represent the best values available. The LISIRD 146 uses for the pre-satellite period 1611-1978 the SATIRE-T TSI dataset with some refine-147 ments included (Kopp et al., 2016); for the satellite period 1979-2018 it used the Community-148 Consensus TSI Composite (Dudok de Wit et al., 2017). Though Kopp's LISIRD data 149 set has no official status, his work as a lead researcher in solar irradiance assessment with 150 satellites is featured with multiple references in IPCC AR5. 151

Because sea surface temperature is being claimed to be unreliable before 1880 due 152 to insufficient data (Smith & Reynolds, 2003) and the ACRIM-PMOD controversy in-153 dicates that there are unsolved problems with TSI data starting from the mid-nineties 154 minimum (Scafetta et al., 2019), the period 1880-1985 is used here as the main research 155 period for studying the solar-climate connection. This choice is also justifiable because 156 prior to 1950 the influence of humans on temperature had been negligible (de Jager et 157 al., 2006); however, the HadSST3 dataset indicates that the rise of sea surface temper-158 ature started in the 2nd half of the 1970s. 159

The data analysis starts with a correlation assessment (based on Pearson correlation coefficient calculated with Excel) for the full data set (1880-2018) and the chosen research period (1880-1985), combined with an assessment focused on the solar minimum years and solar maximum years separately.

Then a temperature profile for the Hale cycle is constructed from the chosen re-164 search period (with 5 successive Hale cycles included). Average values for the two sep-165 arate Hale cycle minima and the two separate Hale cycle maxima serve as reference points 166 in order construct the profile. A linear upward trend is first removed with an improvi-167 sation method (which is necessary due to the irregular length of the solar cycle) using 168 the linear average upward directed trend during the period 1882-1988 in order to make 169 sure that the beginning and ending of the Hale temperature profile show the same value. 170 The slope of the applied trend removal has been checked to be a realistic value that is 171 representative for the temperature rise in the period 1880-1985. The profile is then con-172 structed based on consistent patterns between snippets of the profile found at the sur-173 rounding years near the minima and maxima. 174

Because the minima are known to represent the most stable and most relevant phase 175 of the solar cycle, only the Hale cycle minima are then used to serve for studying the solar-176 climate connection in depth with the introduction of a correction based on the 22-year 177 cycle solar polarity change. The use of a correction based on the 22-year Hale cycle in-178 volves an innovative element that has not been introduced before in reports focused on 179 studying the solar-climate connection. This is initially done for just the minimum years 180 involved, which requires a separation between positive and negative solar minimum years; 181 analyses are made here based on the use of a correlation test combined with an explained 182 variance test (based on R2 method via linear regression analysis + significance levels both 183 calculated with PSPP software). The correction serves to neutralize a structural tem-184 perature difference between the positive and negative solar minimum years. An addi-185 tional analysis is also presented for Hale cycle minima based on multiple years 3 up to 186 187 9 years; an analysis based on 11-year minima is presented as well but it is not taken into consideration for analysis due to overlap between various periods (because for 11-year 188 minima periods some years become included in multiple minima periods - which is ob-189 viously not acceptable). 190

Finally, the SST solar sensitivity is calculated for 3 different perspectives (at the top at the atmosphere, for earth surface after adjusting for the shape of the earth & albedo without an amplification factor, and for earth surface after adjusting for the shape of the earth & albedo with an amplification factor). These 3 perspectives are described for the minima period 1890-1985, for the 22-year cycle and for the 11-year cycle.

The data analysis is available as a spreadsheet. The section 'Electronic Supplemen-196 tary Material' presents online resources available in 2 formats: in (1) Excel format (data 197 + results including calculations) and in (2) CSV format (data + results excluding cal-198 culations). The files are available at a repository download location. The spreadsheet 199 describes for the period 1880-2019 the LISIRD TSI dataset + the HadSST3 dataset +200 all correlations (based on Pearson correlation coefficient) + all explained variances (based 201 on R2 method via linear regression) featured in figures 2 through 5. For the purpose of 202 reproducibility a detailed summary is presented for each of these figures; the data shown 203 in each figure is processed in the data files as follows: 204

- Figure 2: LISIRD TSI (column C), HadSST3 (column D); columns I to AW show data + correlations with regard to the periods 1880-2018 and 1880-1985 for: maxima, odd maxima [O], even maxima [E], minima, positive minima [P], and negative minima [N].
- Figure 3: Temperature profile Hale cycle (column CW), temperature profile Schwabe 209 cycle (column CZ); columns BB to CS present the underlying calculation method 210 for the Hale cycle temperature profile. The Hale cycle temperature profile is com-211 posed of 4 series of data around the TSI minima and maxima years, whereby the 212 profile of the positive minima years is split into 2 parts (column BX and column 213 CN therefore contain the same data). The trend has been removed from each of 214 the 4 reference profiles based on a slope corresponding to a temperature increase 215 of 0,0028 °C per year (= 0.28 °C per 100 years); a higher or lower value would mean 216 that the second positive minimum (for year 22) in figure 3 would not end exactly 217 at zero. Only the values labeled with a * have been processed in the Hale cycle 218 temperature profile. The indicative bandwidths (column DF for the negative min-219 imum, column DH for the temperature peak year) represent the outliers derived 220 from 4 successive full Hale cycles based on the period 1890-1976. The Schwabe 221 cycle temperature profile has been derived from the Hale cycle temperature pro-222 file. 223
- Figure 4: [TOP] LISIRD (column EI), HadSST3 (column EJ); [BOTTOM] LISIRD with corrected negative minima (column EQ), HadSST3 (column ER). The correction value is the lowest value, for which the average correlation value of the positive and negative data combined is found.
- Figure 5: 1-year mean corrected LISIRD (column FR) & HadSST3 (column FS); 228 3-year mean corrected LISIRD (column GQ) & HadSST3 (column GR); 5-year 229 mean corrected LISIRD (column HP) & HadSST3 (column HQ); 7-year mean cor-230 rected LISIRD (IO column) & HadSST3 (IP column); 9-year mean corrected LISIRD 231 (column JN) & HadSST3 (column JO); 11-year mean corrected LISIRD (column 232 KM) & HadSST3 (column KN). The correction value represents the lowest value 233 for each minimum period whereby for the minima combination the average cor-234 relation value of the positive and negative data is found. 235

236 3 Results

With the Hale cycle taken in consideration, correlations between TSI and seawater surface temperature are described first. The period around the minimum years 1890

to 1985 is then used in order to calculate the temperature profile for the 22-year Hale 239 cycle (+ the temperature profile for the 11-year Schwabe cycle). Also, based on the min-240 imum years a description for the solar sensitivity in the long-term perspective is presented. 241 A distinction is made between: (1) 'positive minima' [P] which are formed during the

242

phase with the solar magnetic poles in the original position and (2) 'negative minima' 243

[N] which are formed during the phase when the poles have switched positions. 244



Figure 2. The individual phases of the solar cycle show correlations for the LISIRD TSI total solar irradiance and HadSST3 seawater surface temperature that are significantly higher compared to the values for the entire cycle. The minima show structurally higher correlation values with respect to the maxima. The TSI has a structural impact due to the 22-year magnetic solar cycle, which is expressed in relatively high 'positive TSI minima' [P] and 'odd TSI maxima' [O] (relative to in respective the 'negative TSI minima' [N] and 'even TSI maxima' [E]). This structural phenomenon is in accordance with the Gnevyshev-Ohl rule, which is associated with just the maxima of the sunspot cycle according the literature (Zolotova & Ponyavin, 2015).

245 246

3.1 Total solar irradiance (TSI) & temperature correlate higher during minima than during maxima

Figure 2 describes a stable correlation (r = 0.42) for the TSI and seawater surface 247 temperature showing the same magnitude for both the period 1890-1985 and the period 248 1880-2018. However, for both the minima and maxima of the solar cycle the correlations 249

are at a significantly higher level; in accordance with expectations (Hiyahara et al., 2008),
 the correlation for the individual phases shows the highest level for the minima.

Moreover, correlations at both the positive & negative minima and the odd & even maxima reach an even higher level. For the period 1880-1985, very high correlations with almost the same value are found for both the positive and negative minima. And for the odd and even maxima the same correlation value is found. This indicates that during the course of the 22-year cycle, the fluctuation of the TSI-temperature correlation shows a high degree of regularity.

The structurally higher correlations in the positive and negative minima series (compared to the combination of both series) appear also directly related to the Gnevyshev-Ohl rule; in figure 2 the dashed green curves show the impact for both the TSI minimums and the TSI maximums separately.

262

3.2 Temperature profile for the 22-year & 11-year solar cycle

The HadSST3 seawater surface temperature profile for the 22-year solar cycle has 263 been determined based on the period 1882-1988. This period includes: 5 even maximums, 264 5 positive minimums, 5 odd maximums, and 5 negative minimums. The mean values for 265 these 4 categories serve each as a separate reference point. The average value is then de-266 termined for the years around each of these 4 reference points. This results in 4 refer-267 ence profiles that each show a temperature difference within the range of 0,20-0,27 °C 268 that manifest in 7 to at most 11 years (with an average value of 0,236 °C). The trend 269 has subsequently been removed from each of the 4 reference profiles. Finally, the tem-270 perature profile is compiled by means of the years around the 4 reference points. In par-271 ticular the years around the minimum reference points have been used for this because 272 the years around the two maximum reference points show less consistency compared to 273 the other 2 reference profiles (the method section describes the procedure in detail). The 274 profile for the 11-year Schwabe cycle is derived from the profile of the Hale cycle; only 275 the minima of the Hale temperature profile served as reference points. 276

The temperature profile for the Hale cycle is shown in figure 3. The length of the 277 Hale profile is only 21 years because the Hale cycles in the research period were relatively 278 short: the average length of the Hale cycles in the period 1890-1985 is approximately 21 279 years. For the Hale cycle profile, the largest temperature difference is found between the 280 positive minimum and the phase that follows 2 years before the negative minimum. The 281 (average) temperature difference between the positive minimum and the temperature peak 282 is 0.215 °C. The temperature difference between the positive minimum and the nega-283 tive minimum is 0,059 °C. 284

The TSI negative maximum occurs 4 years after the TSI positive minimum and the TSI even maximum occurs 5 years after the TSI negative minimum. So, the TSI even maximum coincides with the highest temperature value in the 2nd part of the Hale cycle (which starts from the negative minimum and ends at the positive minimum).

Figure 3 shows that the first part and the second part of the Hale cycle show an 289 asymmetrical temperature trend. During the first part, the fluctuations are more frequent 290 and the amplitude is higher relative to the second part. The temperature peaks relatively 291 late in the first part and it peaks relatively early in the second part. In addition, the pro-292 file of the Hale cycle shows an oscillation with fluctuations that take 2 to 7 years, which 293 corresponds to the variation described for the duration of the ENSO cycle. This is not 294 entirely surprising as it is known that there are strong statistical relationships between 295 ENSO and the activity of the Sun (Narsimha & Bhattacharyya, 2010). 296

For the period 1882-1988, the radiative forcing between all adjacent maxima and minima shows an average value of 0,86 W/m2. Combined with the average maximum



Figure 3. Seawater surface temperature profile for the Hale cycle based on the period 1882-1988 (which includes e.g. 5 positive minima and 5 negative minima) shows a maximum impact of 0,215 °C. During the first part of the Hale cycle, the fluctuations are larger than during the second part. The temperature profile for the Schwabe cycle shows a maximum impact for seawater surface temperature of only 0,122 °C. The bandwidths shown for the negative minimum year and temperature peak year (which is found 2 years before the negative minimum year) represent the outliers are derived from 4 successive full Hale cycles based on the period 1890-1976 (solar cycles 13-20) - which involves less data than the temperature profile itself, nevertheless, the 4 values average result is for both phases of the cycle approximately similar to the profile value.

temperature difference within the profile of 0.215 °C this results in a solar sensitivity within 299 the Hale cycle of $0.25 \,^{\circ}\text{C}$ per W/m2 at the top of the atmosphere (TOA); converted to 300 Earth's surface this results in a value of 1,43 °C per W/m2 (via a conversion factor of 301 0,175: 25% based on Earth's spherical formation in combination with 70% albedo). How-302 ever, the impact of an amplifying factor for the TSI signal at the top of the atmosphere 303 has not yet been taken into account in this latter result. In the section discussion & con-304 clusion an amplification factor with a value of 6 is used in order to find the solar sen-305 sitivity on Earth's surface for the Hale cycle, which results in a value of 0.238 °C per W/m2. 306 Likewise, for the 11-year cycle a considerably lower solar sensitivity on Earth's surface 307 is found: 0,135 °C per W/m2. Within the conceptual framework of the IPCC, both the 308 22-year Hale cycle and the amplification factor are being ignored (IPCC, 2013). 309

For the sake of completeness, figure 3 also shows the temperature profile for the 11-year Schwabe cycle (which has been derived directly from the Hale cycle profile). A striking feature of the profile for the Schwabe cycle is that it contains 2 peaks of approximately the same height. This finding is not entirely surprising neither because of the fact that for the 11-year sunspot cycle 2 maxima are also described - which typically arise in a time frame of 2 to 4 years. This period between the two peaks is known as the 'Gnevy-shev gap'; the relative height of the peaks relates to Forbus decreases (which involves rapid decreases in cosmic rays intensity following coronal mass injection) and is indicative for an odd/even cycle (Ahluwalia et al., 2008). In the literature the first sunspots peak relates to UV radiation and the second peak to geomagnetic disturbances (+ aurora phenomena) (Gnevyshev, 1977).

321 322

3.3 Positive & negative TSI minima show high correlation with seawater surface temperature

The upper part of figure 4 describes for the period 1890-1985 a high correlation for 323 TSI and seawater surface temperature with a declared variance of around 90% for both 324 the positive and negative minima. This involves the same correlations that are described 325 for the minima in figure 2; in figure 4 the TSI scale has been adjusted to show the dy-326 327 namics visually. For the positive and negative minima separately, the temperature follows the trend of the TSI (with exception for the first transition of the negative minima 328 where both factors move in opposite directions). However, when the distinction between 329 the positive and negative minima is ignored, 6 out of 9 transitions show an opposite move-330 ment between the TSI and the temperature. This dynamic for the combination is incon-331 332 sistent with the dynamics for the positive and negative minima separately.

The introduction section describes that during the minima of the positive phase 333 temperature typically reaches the lowest level since the Maunder minimum. Regarding 334 the physical mechanism involved it is known that during the negative phase of the so-335 lar cycle the supply of cosmic rays (which is associated with cloud formation (Svensmark, 336 2015)) is more sensitive because then the supply comes more via the equator of the Sun, 337 while during the positive phase the supply comes more via the poles (Hiyahara et al., 338 2008). This implies that based on the direction of cosmic rays supply one can conclude 339 that the relationship between TSI and temperature directly depends on the polarity of 340 the Sun. During the negative phase (which starts at around the odd maximum, during 341 the transition from the positive minimum to the negative minimum) a relatively small 342 amount of energy is needed for a temperature increase, while during the positive phase 343 (which starts around the even maximum, during the transition from the negative min-344 imum to the positive minimum) more energy is required for the same temperature rise. 345 Logically this means that a structural correction is needed to describe (and better un-346 derstand) the relationship between the TSI and the temperature - although the use of 347 a correction is not necessary for a comparison between individual years when these in-348 volve the same phase of the 22 year cycle. 349

In the bottom part of figure 4 a correction has been applied to the negative TSI 350 minimum values. Due to the correction the correlation for the combination of the pos-351 itive and negative minimum values shows the mean value of both minima phases sep-352 arately. This result implicates that in the bottom part of figure 4 the explained variance 353 for the combination ends up at a likewise high percentage (90,5%) as seen for both min-354 ima separately, while in the top part of figure 4 the explained variance is much lower (56,6%). 355 In addition, after using the correction the TSI and temperature move in the same direc-356 tion at all 9 transitions. The solar sensitivity for the combination is 1,20 °C per W/m2 357 at the top of the atmosphere (TOA); converted to Earth's surface this produces a value 358 of $6.86 \,^{\circ}\text{C}$ per W/m². However, this does not yet take into account the influence of the 359 amplifying factor on the TSI signal at the top of the atmosphere; the discussion & con-360 clusion section assumes an amplification value of 6 which results in a solar sensitivity on 361 Earth's surface of: 1,143 °C per W/m2 (= 1,20 / ($(0,25 \ge 0,70) \approx 6$)), which is only slightly 362 lower than the value at the top of the atmosphere. 363

The solar sensitivity of 1,20 °C per W/m2 TOA (for Earth's surface: 1,143 °C per W/m2) for the period 1890-1985 combined with the solar sensitivity during the 22-year



Figure 4. (top) HadSST3 seawater surface temperature plotted against LISIRD TSI (+1360 W/m2) shows that for the period 1890-1985 very high correlations are only found for the positive [P] and negative [N] minima separately; (bottom) after a correction of +0,142 W/m2 focused on the negative TSI values, a very high correlation is also found for the combination of the minima. With the use of a regression analysis, the solar sensitivity at the top of the atmosphere (TOA) for this period is established at: 1,20 °C per W/m2 for the LISIRD TSI values above 1360 W/m2 (based on a declared variance of 90,5%). The values for the minimum year 1912 have been used as reference point.

solar cycle of 0,25 °C per W/m2 TOA (for Earth's surface: 0,238 °C per W/m2) implies that the long-term solar sensitivity is 4,8 times higher than during the short-term perspective of the 22-year cycle. Compared to the 11-year cycle the long-term solar sensitivity is 8,4 times higher.

According the LISIRD TSI dataset, the total solar irradiance between Maunder minimum (1360,274 W/m2 TOA) and the most recent positive minimum year 2017 (1361,215 W/m2 TOA) has increased by 0,941 W/m2 TOA. Based on the long-term solar sensitivity of 1,143 °C per W/m2 after taking into account Earth's shape (25%), albedo (70%) and the amplifying factor (6x) for the TSI signal, this results in a temperature rise at Earth's surface of 1,07 °C (based on the TSI signal of 1,20 °C per W/m2 TOA, the value is slightly higher: 1,13 °C).

For the positive and negative minima separately, the solar sensitivity (TOA) is in respective: 1,10 °C per W/m2 and 1,22 °C per W/m2.

The magnitude of the correction is with a value of more than 0,1 W/m2 about one tenth of the average fluctuation of the TSI during an 11/22 year solar cycle. This represents the same magnitude found at the structural variations of the sunspot cycle based on the Gnevyshev-Ohl rule (Zolotova & Ponyavin, 2015).

383 384

3.4 Multi-year TSI minima show a comparable trend with seawater surface temperature after correction

The correction method aimed at the negative TSI minimum values has also been applied to the minima based on the 3-year, 5-year, 7-year, 9-year and 11-year average values (which involve both sides surrounding the minimum).

Figure 5 shows that the magnitude of the correction for the 3-year to the 9-year average is slightly smaller (0,110-0,138 W/m2) than the correction value for the 1-year minima (0,142 W/m2), but the values show consistently the same order of magnitude. The 11-year average shows an even smaller correction value (0,100 W/m2); however, because there is an overlap between various periods the result for the 11-year period is disregarded in this analysis.

Figure 4 shows for the 1-year to 9-year minima that the first five values of both the LISIRD TSI and the seawater surface temperature are lower than the last five minima. Also, the first five values always show the lowest value at 1912 and the highest value at 1933; for the last five values the year 1976 always shows the lowest value.

Only the 1-year to 5-year minima show the same direction of the trend at all 9 tran-398 sitions for the LISIRD TSI and the HadSST3 seawater surface temperature after apply-399 ing the negative TSI correction. For the 7-year and 9-year minima, eight out of nine tran-400 sitions show the same trend direction; only the transition between 1943 and 1954 shows 401 opposite trends. Figure 2 presents an explanation for this exception because the 1958 402 maximum (+ the immediately surrounding years) is the largest outlier in the LISIRD 403 TSI dataset. This phenomenon also explains why in figure 5 the highest average TSI value is found at the 1954 minimum for both the 7-year and 9-year average, while the 1-year 405 to 5-year show the highest level for both the TSI and the temperature at the 1943-value. 406

For the 1-year to 9-year minima, the explained variance is within the bandwidth of 89.2-92.8% after applying the correction aimed at the negative minima. With increasing length of the minima periods the value of the explained variance fluctuates only a few percent from the 90,5% explained variance found at the 1-year minima for the postive and negative minima separately, as well as for the combination of both minima including the correction. Correlation significance levels show that the results are highly significant [p=0,000] for all minima periods shown in figure 5.

When the correction value based on the 1-year minima period (0,142 W/m2) would have been applied to all other perspectives, only the explained variance for the 3-year



Figure 5. After applying a correction aimed at the negative minima, the 1-year, 3-year, 5-year, 7-year, and 9-year periods around the minima show similar dynamics. The first 5 values of both the LISIRD TSI and the HadSST3 are below the last 5 values. For the first 5 values the 1912 minimum always shows the lowest value and the 1933 minimum shows the highest value; for the last 5 values the 1976 minimum always shows the lowest value.

period would show a small drop (from 92,5% to 92,4%). The 5-year to 9-year periods
would then show a further rise for the explained variance.

4 Discussion & Conclusions

This article investigates the Sun's impact on climate with the 22-year magnetic solar cycle. The solar sensitivity is described in 3 forms: (1) in terms of the TSI at the top of the atmosphere; (2) this value is then converted to Earth's surface via a correction for the spherical Earth (25%) and the albedo factor (70%); (3) finally, it has also been corrected with an amplifying factor which increases the temperature impact of the TSI signal at the top of the atmosphere.

For a calculation of the temperature impact of the Sun over a certain period, it is not strictly necessary to make the conversion to Earth's surface when phase differences within the 22-year cycle are taken into account. However, this conversion does become necessary for a description of the solar sensitivity on Earth's surface in terms of the radiative forcing. Therefore, the impact of the amplification factor will now be discussed in more detail here (without going into the possible physical mechanisms involved).

Since the 1990s experts have speculated about the impact of an amplifying factor
for the TSI signal formed by the Sun at the top of the atmosphere. Literature has taken
into account the possibility that the magnitude of the amplification factor could theoretically vary at the order of 2 to 10 times (Stott et al., 2003). However, there is no consensus about the exact magnitude; therefore, controversy also exists on this matter. Estimates appear to depend, among other things, on the TSI dataset used (Haigh, 2007).

Based on 20th century data, the estimates range from 2-3 times (Haigh, 2007), 3 437 times (Stott et al., 2003), 4-6 times (Ziskin & Shaviv, 2012) up to as high as 4-8 times 438 (Holmes, 2018). The IPCC confirms that there is great uncertainty about the radiative 439 forcing of the Sun (Haigh, 2007). The most detailed estimates have been described based 440 on the 11-year solar cycle, where the values for the amplification factor are relatively high: 441 5-7 times (Shaviv, 2008). As far as is known, there are no descriptions which indicate 442 that there are concrete reasons to assume that the magnitude of the amplifying factor 443 444 for the TSI signal also fluctuates. Therefore, it is assumed here that there is a stable amplification factor with a value of 6 combined with a bandwidth of 5 to 7. 445

This implies that the Sun's sensitivity at Earth's surface is (only) slightly lower com-446 pared to the value measured at the top of the atmosphere. After all factors have been 447 taken into account, the result via the chosen amplifying value (6 times) amounts to 95%448 of the TOA value. If the amplification value were slightly lower, then the Earth's sur-449 face would have almost the same value as the TSI at the top of the atmosphere (with 450 an amplification value of 5,7 times it would produce almost exactly the same value). The 451 bandwidth for the amplifying factor is used here to describe an indication for the un-452 certainty margin of the solar sensitivity specific to the perspective of Earth's surface af-453 ter all factors have been taken into account. 454

455 For the three perspectives examined, the following values are found in regard to 456 solar sensitivity:

457	• 11-year cycle:
458	- Solar sensitivity based on just TSI at top of atmosphere [TOA]: 0,142 $^{\circ}\mathrm{C}$ per W/m2.
459	- Solar sensitivity converted to surface without amplifying factor: 0,81 $^{\circ}\mathrm{C}$ per W/m2.
460	- Solar sensitivity converted to surface with amplifying factor (5-7 times): $0,135$
461	\pm 0,03 °C per W/m2.

• 22-year cycle:

- Solar sensitivity based on just TSI at top of atmosphere [TOA]: 0,25 °C per W/m2.
 Solar sensitivity converted to surface without amplifying factor: 1,43 °C per W/m2.
 Solar sensitivity converted to surface with amplifying factor (5-7 times): 0,238
 ± 0,05 °C per W/m2.
- Period 1890-1985:

468

469

470

471

- Solar sensitivity based on just TSI at top of atmosphere [TOA]: 1,20 °C per W/m2.
- Solar sensitivity converted to surface without amplifying factor: 6,86 °C per W/m2.
 - Solar sensitivity converted to surface with amplifying factor (5-7 times): 1,143
- \pm 0,23 °C per W/m2.

This overview shows that the solar sensitivity at Earth's surface depends especially on the magnitude of the amplification factor. The value of the solar sensitivity at Earth's surface increases when the amplifying factor decreases. This also applies to the albedo factor because a lower albedo value leads to a higher result in the calculation of the solar sensitivity for Earth's surface.

This implies that solar sensitivity for the long-term perspective is more than 4 times 477 (4,8 times) higher than during the short-term perspective of the 22-year magnetic so-478 lar cycle; when compared with the perspective of the 11-year sunspot cycle, the value 479 for the long-term perspective is more than 8 times (8.4 times) higher. These values are 480 approximately 2 times higher than the ratios described in literature relative to the 11-481 year solar cycle (de Jager et al., 2006; Shaviv, 2005, 2012). These results also confirm 482 earlier descriptions based on periods that go further back in time, which show that the 483 temperature impact during the 22-year cycle is much larger (here 78%) than during the 484 11-year cycle; in a study by Scafetta & West (Scafetta, 2005) a 54% higher value is re-485 ported for the 22-year cycle $(0.17 \pm 0.06 \text{ °C per W/m2})$ versus the 11-year cycle (0.11 cycle for a cycle)486 ± 0.02 °C per W/m2). Related literature also confirms that the change of magnetic po-487 larity plays a key role in this (Hiyahara et al., 2008). 488

The IPCC describes in AR5 (2013) a temperature effect for the 11-year cycle with fluctuations at the order of 0,03-0,07 °C (mean value 0,05 °C) (IPCC, 2013); the temperature profile for the 11-year cycle in figure 3 shows fluctuations with an average value of 0,122 °C which is more than 2 times higher than the IPCC description.

Based on long-term solar sensitivity, it has been calculated that the Sun can be held 493 responsible for a temperature rise of approximately 1,1 °C since Maunder minimum (late 494 17th century). Estimates for the total warming since Maunder minimum are in the or-495 der of 1,5 °C (PAGES2k Consortium, 2019). Estimates for the temperature difference 496 between a passive and active Sun are in the order of 1 $^{\circ}C$ (Shaviv, 2012) (up to 2 $^{\circ}C$). 497 Since the start of the Holocene 11,700 years ago, the activity of the Sun has shown the 498 highest change between Maunder minimum and the early 21st century (Usoskin et al., 499 2007). An estimate is also available which describes that the increase in solar activity 500 since the emergence of life on Earth can explain about half to 2/3 of the temperature 501 increase (Karoff & Svensmark, 2010; Scafetta, 2013). These estimates are consistent with 502 the long-term solar sensitivity described here based on the period 1890-1985. 503

Because the solar minimum years do not coincide with the start and end of the 20th 504 century, it is not possible to make an exact calculation based on the minima for the share 505 of the Sun in the seawater surface temperature rise between 1900 and 2000, which is about 506 0,416 °C. However, an indicative calculation can be made on the basis of the negative 507 minima in the period 1902-2008 (this period covers almost the entire 20th century). For 508 the proportion of the Sun, the percentage here amounts to 62,1% of the 0,671 °C warm-509 ing of the seawater surface temperature between 1902 and 2008; this percentage is not 510 far below the upper limit of 69% described by Scafetta & West for the period 1900-2005 511 (Scafetta & West, 2008). For the period 1890-2017 the Sun provides a share of 58.2%512 in the warming of 0,928 °C. Both percentages are around 60% - just below the upper 513

limit of 64% of the bandwidth described in the introduction for the global warming in
the 20th century (Scafetta, 2013).

For the 21st century, a comparison between the positive minimum years 1996 and 2017 provides a remarkable picture, because based on the solar sensitivity of 1,2 °C per W/m2 the entire temperature rise (103.6%) is explained by the Sun. However, a comparison between the positive minimum years 1954 and 2017 yields a percentage of the sun that is less than half (46.4%).

From an energetic point of view, the solar sensitivity for the long-term perspective 521 at Earth's surface (with the amplification factor included) shows with a value of 1,143 522 ± 0.23 °C per W/m2 a measure for the equilibrium climate sensitivity parameter (λ). 523 The temperature impact of this is comparable to a climate sensitivity for the doubling 524 of CO2 with a bandwidth of 3,38-5,08 °C (based on: 3,7 W/m2 x 1,143 \pm 0,23 °C per 525 W/m2). The midpoint of this bandwidth is found at the value 4,23 °C, which is below 526 the upper limit of the bandwidth that the IPCC applies for climate sensitivity: 1,5-4,5 527 $^{\circ}$ C (IPCC, 2013). An additional comment follows based on the period 1912-1965. 528

Based on the dynamics in the lower part of figure 4, the period 1912-1965 shows 529 an almost perfect correlation (combined with an explained variance of 99%) between the 530 minimum values of LISIRD TSI and HadSST3 seawater surface temperature. If the cal-531 culation had been made on the basis of the period 1912-1965, the solar sensitivity would 532 drop from 1,20 °C per W/m2 to 1,05 °C per W/m2 (with the use of an unchanged cor-533 rection aimed at the negative minima of 0.142 W/m2). The warming after the Maun-534 der minimum would then amount to 0,99 °C based on the period 1912-1965 and the so-535 lar sensitivity would amount to 1.00 ± 0.20 °C per W/m2 based on the amplifying fac-536 tor (6x). This is energetically comparable to a climate sensitivity for doubling CO2 with 537 a bandwidth of 2,96-4,44 °C. This bandwidth corresponds to the upper side of the IPCC 538 bandwidth. The explained variance of 99% for the 53-year period 1912-1965 offers hardly 539 any impact for influences other than the Sun. This suggests that the Sun is most likely 540 responsible for the temperature trend at least until 1965. Based on the period 1912-1965 541 the solar sensitivity for the long-term perspective is 4,2 times higher than the short-term 542 perspective of the 22-year cycle and 7,4 times higher than the short-term perspective of 543 the 11-year cycle. 544

The correction shows that there is an opposite temperature effect present around 545 the phenomenon related to the Gnevyshev-Ohl rule. Moreover, the phenomenon itself 546 applies to both the TSI maximums and the TSI minimums in the full period starting from 547 1880 (see figure 2). The SATIRE-T TSI dataset (Wu et al., 2018) shows that the rel-548 atively low negative TSI minima involve a pattern which origins from the ER magnetic 549 flux (ephemeral regions); the pattern is not present in the AR magnetic flux (active re-550 gions - which strongly correlate with sunspots and F10.7 radio flux |http://lasp.colorado 551 .edu/lisird/data/noaa_radio_flux/ (NOAA, 2018)]), nor in the open magnetic flux 552 (coronal source flux). ER magnetic flux is missing in early TSI reconstruction methods 553 (Lean et al., 1995; Hoyt & Schatten, 1993), which explains why the pattern is not present 554 in those datasets. The SATIRE dataset also serves for CMIP6 modellers (Matthes et al., 555 2017). The top of figure 6 displays the sunspot cycle (which shows the 11-year period-556 icity of the Schwabe cycle); the bottom of figure 6 displays the cosmic ray flux, which 557 shows an 22-year alternating pattern of flat [qA>0] and peaked [qA<0] tops that coin-558 cides with the solar minima of the sunspots cycle. 559

The magnitude of the correction appears to be more or less independent of the length of the minimum period used in the calculation; the bandwidth of the correction ranges from 0,110-0,148 W/m2 for the values based on 1 to 9 year periods around the TSI minima. This means that there is a structural temperature effect that, in terms of magnitude, approximately corresponds to the average impact of the fluctuations based on the Gnevyshev-Ohl rule. The direction of the temperature effect can be explained on the ba-

sis of a sensitivity difference for the influence of cosmic rays during the positive and neg-566 ative phase of the Hale cycle (Hiyahara et al., 2008). During the negative phase, the cli-567 mate is more sensitive to the supply of cosmic rays than during the positive phase. The 568 negative minimum falls in the middle of the negative phase (see figure 1). As a result 569 the influence of the loss of cosmic radiation due to the poloidal maximum is relatively 570 large, which results in relatively high temperatures during the negative TSI minima. Both 571 the mechanism involved with this temperature effect (as a result of the change of the mag-572 netic solar poles), as well as the magnitude of the associated impact of the temperature 573 effect (comparable with the impact of the Gnevyshev-Ohl rule) have been identified by 574 approximate. 575



Figure 6. The 11-year periodicity of the Schwabe cycle based on sunspots (top); the 22-periodicity of cosmix rays flux indicated by flat [qA>0] and peaked [qA<0] tops (bottom) (Ross & Chaplin, 2019).

The temperature development might be directly related to background solar irradiance [BSI], which concerns the radiation of the Sun excluding the influence of solar flares and sunspots. BSI involves a dynamic component on top of the base level in the signal from the Sun measured at the top of the atmosphere. Uncertainty margins for the baseline (which itself is estimated at around 1361 W/m2 since 2008) are significantly lower than for the TSI fluctuations which arise from magnetic activity due to solar flares $[T_F]$ and sunspots $[T_S]$. This might also explain why the correlation between sunspots and temperature is low; for, both do not involve the background component at all. Equation (1) (Lean et al., 1995) defines that TSI [T(t)] represents the sum of different components. Equation (1) contains only 2 magnetic components, which is in accordance with the Lean method (Coddington et al., 2016); however, a dynamic BSI component that fluctuates over time on top of the base level component $[T_Q]$ is missing:

$$T(t) = T_Q + \Delta T_F(t) + \Delta T_S(t) \tag{1}$$

For the period 1890-1985, the LISIRD TSI dataset shows high correlations with 588 the NRLTSI2 dataset (0.903), IPCC AR5 dataset (0.938) and Satire S&T dataset (0.944). 589 Correlations among the other 3 TSI datasets fall within the bandwidth 0,927-0,998. For 590 the period 1985-2012, the LISIRD TSI dataset shows a high correlation with the NRLTSI2 591 dataset (0,961) but lower correlations are found with the IPCC AR5 dataset (0,846) and 592 Satire S&T dataset (0,868). For this period correlations among the other 3 TSI datasets 593 fall within the bandwidth 0,941-0,984. For the entire period 1890-2012, the LISIRD also 594 shows comparable correlations with the other datasets (0.916-0.926); correlations among 595 the other 3 TSI datasets fall within the bandwidth 0,925-0,995. The period until the year 596 2012 has been considered here because the IPCC AR5 TSI dataset ends in the year 2012. 597

The LISIRD dataset shows for the satellite era a continuous upward trend for the TSI minima since the mid-eighties. A similar continuous upward trend for the TSI minima in the satellite era is described by the Belgian RMIB TSI dataset (DeWitte & Nevens, 2016). The authors of both datasets are involved with the Community-Consensus TSI composite, which also shows this trend.

Here the conclusion is made that the Sun is responsible for the formation of an climate oscillation with an upward slope. With consideration of the 22-year TSI cycle, the high explained variances with a bandwidth of 89-93% for the various minimum periods around the period 1890-1985 (99% for the 1912-1965 minima) leave little room for a large influence of other factors, such as CO2. However, when the 22-year cycle is ignored, it is not possible to notice (nor to describe) this strong relationship between solar activity and temperature.

The IPCC climate models do not take into account temperature effects that arise 610 as a result of: (1) the changes of the magnetic solar poles within the 22-year cycle; the 611 same applies to (2) the influence of an amplifying factor on the impact of the TSI sig-612 nal at the top of the atmosphere. Climate models also do not take into account the dy-613 namics that ensure that (3) the solar sensitivity within the 11-year TSI cycle is signif-614 icantly lower than in the multi-decadal long-term perspective. In determining short-term 615 trends, climate models neither take into account (4) the impact of the upward phase of 616 the multi-decadal cycle, which can be directly connected with the Gleissberg cycle min-617 ima of the Sun (Feynman & Ruzmaikin, 2014), nor do climate models consider the in-618 fluence of very long-term solar related cycles such as for example: Jose cycle 179 years 619 (Jose, 1965), de Vries/Suess cycle 248 years (Holmes, 2018), Eddy cycle 1000 years (Holmes, 620 2018), and Hallstatt cycle 2400 years (Usoskin et al., 2016) / Bray cycle 2500 years (Holmes, 621 2018). The missing of this set of 4 solar-related factors in climate models points towards 622 a significant structural underestimation of the Sun's impact on the climate, leading to 623 an overestimation of the impact of CO2 and other natural greenhouse gases. ENSO and 624 NAO represent two other factors (next to e.g. greenhouse gasses) which are known in-625 fluence sea surface temperature; however, both factors also show high correlations with 626 solar activity (Kirov & Georgieva, 2002). Fundamentally it is important that the great-627 est temperature effects due to the change of the magnetic poles can be expected around 628 the solar minima, because during these periods the magnitude of the poloidal magnetic 629 field reaches the highest magnitude - see figure 1. Finally, one side note is made here: 630 for, the influence of mankind on the climate system has become evident particularly through 631

ozone layer depletion resulting from the use of artificial greenhouse gases (especially CFCs);

- despite the relatively large influence of the Sun, the impact of anthropogenic influences
- ⁶³⁴ must therefore be acknowledged.

⁶³⁵ The following abbreviations are used in this manuscript:

ACRIM	Active Cavity Radiometer Irradiance Monitor Satellite
HadSST	Hadley Centre Sea Surface Temperature
IPCC AR5	Intern. Panel on Climate Change Ass. Report 5 (2013)
KNMI	Koninklijk Nederlands Meteorologisch Instituut
LISIRD	Las Interactive Solar Irradiance Data Center
NIOZ	Nederlands Instituut voor Onderzoek der Zee
NRLTSI	Naval Research Laboratory Total Solar Irradiance
PMOD	Physikalisch Meteorologisches Observatorium Davos
SATIRE	Spectral And Total Irradiance REconstructions
TSI	Total Solar Irradiance
WSO	Wilcox Solar Observatory

637 Acknowledgments

- The author gratefully acknowledges the efforts of various informal reviewers from The
- ⁶³⁹ Netherlands who provided much useful advice.

640 Funding:

636

⁶⁴¹ This research received no funding.

⁶⁴² Disclosure of Potential Conflicts of Interest:

⁶⁴³ The author declares no conflict of interest.

644 Electronic Supplementary Material:

- ⁶⁴⁵ The data analysis is available as a spreadsheet in:
- Excel format (Manuscript-datasheet-Excel.xlsx file: data with calculations).
- CSV format (Manuscript-datasheet-CSV.csv file: data without calculations).
- Files are available at url: https://osf.io/qk9sm/files/

649 References

Ahluwalia, H. S., Kritchfield, K., Ygbuhay, R., & Kamide, Y. (2008).Gnevvshev 650 gap in the annual frequency distribution of cosmic ray decreases and Ap index. 651 In R. Caballero, J. C. D'Olivo, G. Medina-Tanco, L. Nellen, F. A. Sanchez, 652 & J. F. Valdes-Galicia (Eds.), Proceedings of the 30th international cos-653 mic ray conference (p. 385-388). Mexico City: World Scientific Publishing 654 Co. Pte. Ltd. Retrieved from https://www.researchgate.net/profile/ 655 Roger_Ygbuhay/publication/252722756_Gnevyshev_gap_in_the_annual 656 _frequency_distribution_of_cosmic_ray_decreases_and_Ap_index/links/ 657 53d325950cf220632f3cc7f1/Gnevyshev-gap-in-the-annual-frequency 658 -distribution-of-cosmic-ray-decreases-and-Ap-index.pdf 659 Camp, C. D., & Tung, K. K. (2007). Surface warming by the solar cycle as revealed 660 by the composite mean difference projection. Geophys. Res. Lett., 34, L14703. 661 doi: https://doi.org/10.1029/2007gl030207 662

663	Cheng, L., Zhu, J., Abraham, J., Trenberth, K. E., Fasullo, J. T., Zhang, B.,
664	Song, X. (2019). 2018 Continues Record Global Ocean Warming. Adv. Atmos.
665	Sci., 36, 249-252. doi: https://doi.org/10.1007/s00376-019-8276-x
666	Coddington, O., Lean, J. L., Pilewskie, P., Snow, M., & Lindholm, D. (2016). A
667	solar irradiance climate data record. Bull. Amer. Meteor. Soc., 97, 1265-1282.
668	doi: https://dx.doi.org/10.1175/bams-d-14-00265.1
669	de Jager, C., Versteegh, G. M., & van Dorland, R. (2006). Climate Change Scien-
670	tific Assessment and Policy Analysis - Scientific Assessment of Solar Induced
671	Climate Change (R. van Dorland, Ed.). de Bilt: Netherlands Environmental
672	Assessment Agency. Retrieved from https://www.pbl.nl/sites/default/
673	files/downloads/500102001_0.pdf
674	DeWitte, S., & Nevens, S. (2016). The total solar irradiance climate data
675	record. The Astrophysical Journal, 830, 25. doi: http://dx.doi.org/10.3847/
676	0004-637X/830/1/25
677	Dudok de Wit, T., Kopp, G., Frölich, C., & Schöll, M. (2017). Methodol-
678	ogy to create a new total solar irradiance record: Making a composite
679	out of multiple data records. Geophys. Res. Lett., 44, 1196-1203. doi:
680	https://doi.org/10.1002/2016GL071866
681	Feynman, J., & Ruzmaikin, A. (2014). The Centennial Gleissberg Cycle and its
682	association with extended minima. J. Geophys. ResSPACE, 119, 6027-6041.
683	doi: https://dx.doi.org/10.1002/2013jA019478
684	Gnevyshev, M. N. (1977). Essential features of the 11-year solar cycle. Sol. Phys.
685	51, 175-183. Retrieved from http://adsabs.harvard.edu/full/1977SoPh
686	.51175G
687	Haigh, J. D. (2007). The Sun and the Earth's Climate. Living Rev. Sol. Phys., 4, 1-
688	64. doi: https://dx.doi.org/10.12942/lrsp-2007-2
689	Hale, G. E. (1908). On the probable existence of a magnetic field in sun-spots. As-
690	trophys. J., 28, 315-343. Retrieved from http://articles.adsabs.harvard
691	.edu/full/1908ApJ28315H/0000315.000.html
692	Hivahara, H., Yokovama, Y., & Masuda, K. (2008). Possible link between multi-
693	decadal climate cycles and periodic reversals of solar magnetic field polar-
694	ity. Earth Planet. Sci. Lett., 272, 290-295. doi: https://doi.org/10.1016/
695	j.epsl.2008.04.050
696	Holmes, R. I. (2018). Thermal Enhancement on Planetary Bodies and the Rele-
697	vance of the Molar Mass Version of the Ideal Gas Law to the Null Hypothesis
698	of Climate Change. Earth Sci., 7, 107-123. doi: https://doi.org/10.11648/
699	j.earth.20180703.13
700	Hoyt, D. V., & Schatten, K. H. (1993). A Discussion of Plausible Solar Irradiance
701	Variations, 1700-1992. J. Geophys. ResSPACE, 98, 18,895-18,906. doi:
702	https://doi.org/10.1029/93ja01944
703	IPCC. (2013). Climate Change 2013: The Physical Science Basis. Contribu-
704	tion of Working Group I to the Fifth Assessment Report of the Intergovern-
705	mental Panel on Climate Change (T. F. Stocker et al., Eds.). Cambridge,
706	United Kingdom and New York, NY, USA: Cambridge University Press.
707	Retrieved from https://www.ipcc.ch/site/assets/uploads/2018/02/
708	WG1AR5_all_final.pdf (Citations: page 56 (Technical Summary): "Longer
709	term forcing is typically estimated by comparison of solar minima (during
710	which variability is least)." Citation page 689 (Chapter 8): "The year 1750,
711	which is used as the preindustrial reference for estimating RF, corresponds to
712	a maximum of the 11-year SC. Trend analysis are usually performed over the
713	minima of the solar cycles that are more stable. For such trend estimates, it
714	is then better to use the closest SC minimum, which is in 1745 Maxima to
715	maxima RF give a higher estimate than minima to minima RF, but the latter
716	is more relevant for changes in solar activity.")
717	Jose, P. D. (1965). Sun's motion and sunspots. Astron. J., 70, 193-200. doi:

718	https://dx.doi.org/10.1086/109714
719	Karoff, C., & Svensmark, H. (2010). How did the Sun affect the climate when life
720	evolved on the Earth? - A case study on the young solar twin $\kappa 1$ Ceti. $arXiv$
721	preprint, 1-5. Retrieved from https://arxiv.org/pdf/1003.6043.pdf
722	Kirov, B., & Georgieva, K. (2002). Long-term variations and interrelations of ENSO,
723	NAO and solar activity. Phys. Gem. Earth, 27, 441-448. doi: https://doi.org/
724	10.1016/S1474-7065(02)00024-4
725	Kopp, G. (2019). Historical total solar irradiance reconstruction, time series. Re-
726	trieved from http://lasp.colorado.edu/lisird/data/historical_tsi/
727	Kopp, G., Krivova, N., Wu, C. J., & Lean, J. (2016). The Impact of the Revised
728	Sunspot Record on Solar Irradiance Reconstructions. Sol. Phys., 291, 2951-
729	2965. doi: https://doi.org/10.1007/s11207-016-0853-x
730	Lean, J., Beer, J., & Bradley, R. (1995). Reconstruction of solar irradiance since
731	1610: Implications for climate change. Geophys. Res. Lett., 22, 3195-3198. doi:
732	https://doi.org/10.1029/95gl03093
733	Matthes, K., Funke, B., Andersson, M., Barnard, L., Beer, J., & Charbonneau, P.
734	(2017). Solar forcing for CMIP6 (v3.2). Geosci. Model Dev., 10, 2247-2302.
735	doi: https://doi.org/10.5194/gmd-10-2247-2017
736	MetOffice. (2020). Met office hadley centre observations datasets: time series
737	(annual globe). Retrieved from https://www.metoffice.gov.uk/hadobs/
738	hadsst3/data/download.html
739	Mursula, K., & Hiltula, T. (2003). Bashful ballerina: Southward shifted heliospheric
740	current sheet. Geophys. Res. Lett., 30, 2135. doi: https://doi.org/10.1029/
741	2003GL018201
742	Narsimha, R., & Bhattacharyya, S. (2010). A wavelet cross-spectral analysis of
743	solar-ENSO-rainfall connections in the Indian monsoons. Appl. Comput. Har-
744	mon. Anal., 28, 285-295. doi: https://doi.org/10.1016/j.acha.2010.02.005
745	NOAA. (2018). Noaa adjusted solar radio flux at 10.7cm, time series. Retrieved
746	from http://lasp.colorado.edu/lisird/data/noaa_radio_flux/
747	PAGES2k Consortium. (2019). Consistent multi-decadal variability in global
748	temperature reconstructions and simulations over the Common Era (Supple-
749	mentary Materials). Nat Geosci., 12, 643-649. doi: https://doi.org/10.1038/
750	s41561-019-0400-0
751	Ross, E., & Chaplin, W. (2019). The Behaviour of Galactic Cosmic-Ray Intensity
752	During Solar Activity Cycle 24. Sol. Phys., 294, 8. doi: https://doi.org/10
753	.1007/s11207-019-1397-7
754	Scafetta, N. (2005). Estimated solar contribution to the global surface warming us-
755	ing the ACRIM TSI satellite composite. Geophys. Res. Lett., 32, L18713. doi:
756	https://dx.doi.org/10.1029/2005gl023849
757	Scafetta, N. (2013). Discussion on common errors in analyzing sea level accelera-
758	tions, solar trends and global warming. Pattern. Recogn. Phys., 1, 37-57. doi:
759	https://doi.org/10.5194/prp-1-37-2013
760	Scafetta, N., & West, B. J. (2008). Is climate sensitive to solar variability? Phys.
761	Today, 61, 50-51. doi: https://dx.doi.org/10.1063/1.2897951
762	Scafetta, N., Willson, R. C., Lee, J. N., & Wu, D. L. (2019). Modeling Quiet Solar
763	Luminosity Variability from TSI Satellite Measurements and Proxy Models
764	during 1980-2018. Remote Sens., 11, 2569. doi: https://doi.org/10.3390/
765	rs11212569
766	Shaviv, N. J. (2005). On climate response to changes in the cosmic ray flux and ra-
767	diative budget. J. Geophys. ResSPACE, 110, A08105. doi: https://doi.org/
768	10.1029/2004ja010866
769	Shaviv, N. J. (2008). Using the oceans as a calorimeter to quantify the solar radia-
770	tive forcing. J. Geophys. ResSPACE, 113, A11101. doi: https://doi.org/10
771	.1029/2007JA012989
772	Shaviv, N. J. (2012). The Role of the Solar Forcing in the 20th century climate

773	change. In A. Zichichi & R. Ragaini (Eds.), International seminar on nuclear
774	war and planetary emergencies - 44th session (p. 279-286). Singapore: World
775	Scientific Publishing Co. Pte. Ltd. Retrieved from http://citeseerx.ist
776	.psu.edu/viewdoc/download?doi=10.1.1.708.9707&rep=rep1&type=pdf
777	Smith, T. M., & Reynolds, R. W. (2003). Extended Reconstruction of Global Sea
778	Surface Temperatures Based on COADS Data (1854-1997). J. Clim., 16, 1495-
779	1510. doi: https://doi.org/10.1175/1520-0442-16.10.1495
780	Solanki, S. K., Krivova, N. A., & Haigh, J. D. (2013). Solar Irradiance Variability
781	and Climate. Annu. Rev. Astron. Phys., 51, 311-351. doi: https://doi.org/10
782	.1146/annurev-astro-082812-141007
783	Stott, P. A., Jones, G. S., & Mitchell, J. B. (2003). Do Models Underestimate the
784	Solar Contribution to Recent Climate Change? J. Clim., 16, 4079-4093. doi:
785	https://doi.org/10.1175/1520-0442(2003)016(4079:dmutsc)2.0.co;2
786	Svensmark, H. (2015). Cosmic rays, clouds and climate. Europhys. News, 46, 26-29.
787	doi: https://doi.org/10.1051/epn/2015204
788	Usoskin, I. G., Gallet, Y., Lopes, F., Kovaltsov, G. A., & Hulot, G. (2016). So-
789	lar activity during the Holocene: the Hallstatt cycle and its consequence
790	for grand minima and maxima. Astron. Astrophys., 587, A150. doi:
791	https://doi.org/10.1051/0004-6361/201527295
792	Usoskin, I. G., Solanki, S. K., & Kovaltsov, G. A. (2007). Grand minima and max-
793	ima of solar activity: new observational constraints. Astron. Astrophys., 471,
794	301-309. doi: https://dx.doi.org/10.1051/0004-6361:20077704
795	Wu, C. J., Krivova, N., Solanki, S., & Usoskin, I. (2018). Solar total and spec-
796	tral irradiance reconstruction over the last 9000 years. Astron. Astrophys., 620,
797	A120. doi: https://doi.org/10.1051/0004-6361/201832956
798	Ziskin, S., & Shaviv, N. J. (2012). Quantifying the role of solar radiative forcing
799	over the 20th century. Adv. Space Res., 50, 762-776. doi: https://dx.doi.org/
800	10.1016/j.asr.2011.10.009
801	Zolotova, N. V., & Ponyavin, D. I. (2015). The Gnevyshev-Ohl Rule and Its
802	Violations. Geomagn. Aeron., 55, 902-906. doi: https://doi.org/10.1134/
803	s0016793215070300